# Getting clear on corporate culture: conceptualisation, measurement and operationalisation

# NIEN-HÊ HSIEH, BENJAMIN LANGE, DAVID RODIN and MIRA L.A. WOLF-BAUWENS

Abstract: This article provides a review of existing literature on corporate culture, drawing on work from the disciplines of business ethics, management studies, psychology, anthropology, and economics, as well as interviews with business leaders. It surveys different definitions of corporate culture and proposes a framework for capturing their commonalities. It then discusses the importance of culture so conceived and widely used frameworks to measure it. The article also portrays different views on how culture can be operationalised and moulded within an organisation. The article concludes by discussing the relationship between corporate culture and corporate purpose and highlighting gaps in the literature which would profit from further research.

*Keywords:* Corporate culture, organisational culture, culture change, business ethics, corporate purpose, corporate culture significance, culture measurements.

#### **EXECUTIVE SUMMARY**

The construct of organisational or corporate culture is elusive and so there is disagreement and confusion as to its precise definition, importance, and measurement.

This paper reviews more than seventy articles on corporate culture, drawing on work from the disciplines of business ethics, management studies, psychology, anthropology, and economics, as well as interviews with twenty-four business leaders, to offer an integrative and holistic review of corporate culture.

It explores notions of culture and the frameworks most often used to measure it. The authors set out different views on how culture can be operationalised and moulded within an organisation and consider the relationship between corporate culture and corporate purpose.

Organisational culture research emerged in the 1970s from anthropology and sociology, which used qualitative methodologies to focus on how values develop and

are transmitted within groups. That research had its roots even further back in the Gestalt psychology of the 1930s, which introduced the idea of 'social climate'.

The paper considers the constructs of 'culture' and 'social climate' as largely congruent. It surveys a large variety of different definitions and captures their commonalities: organisational culture is a multi-layered, scalar, social phenomenon, concerned with values and related to actions.

Culture is often seen as the 'soft' side of business, but the literature shows it is critical for the successful implementation of strategy, for business performance, and how the corporation operates in a socio-political context. When aligned with personal values, drives, and needs, culture can unleash tremendous amounts of energy. On the other hand, a 'false' culture, misaligned with corporate strategy and purpose, can inhibit changes.

Culture is also important in relation to politics. Governments can manipulate corporate culture to further their own agenda or consolidate their grip on power. What is not yet clear is whether there is a correlation, and the direction of influence between politics and corporate culture.

Measurement frameworks of organisational culture may use quantitative or qualitative measures, or a combination of both. Examples of some of these frameworks, their commonalities, and differences, are given. However, the authors note there is little consistency amongst different methodologies, and no pre-eminent approach.

To be more than empty words, culture needs to be embedded in any organisation's practices. There are many ways of doing this, including 'innovation parenting' to encourage employees to internalise the values of the company. Most approaches rely on the organisation's leaders to live the core values and strategic priorities of the culture, ensuring flat hierarchies and avoiding micromanagement.

The paper identifies how culture affects individual behaviour both within and outside the workplace, how it can help or hinder corporate innovation, and how it can be affected by location, office design, and staff diversity, with examples from other industries, including technology, engineering, and consumer goods.

The authors conclude that culture is only one ingredient in the recipe that leads to a (good) corporation of the future. It is connected to many other areas, most importantly the question of a corporation's purpose, where there is a significant gap in understanding the connection between social purpose and culture.

There is also a need for closer examination of the conditions that influence the different measurement frameworks of culture, for a more holistic study of approaches to operationalise culture and the role of leadership in those efforts.

#### 1 INTRODUCTION

The construct of *organisational* or—as we shall refer to it throughout this review—*corporate* culture is elusive. While there seems to be consensus that organisations of varying size can have 'culture' and that culture is somehow important for an organisation, there exists a lack of clarity and disagreement as to how the construct is to be defined, what drives its importance, and how it can be consequently measured and influenced.

The aim of this article is to offer a review of the literature on corporate culture. It is structured along the above dimensions. We not only draw on the insights of more than seventy articles from the fields of business ethics, management studies, psychology, anthropology, and economics, but also on interviews with twenty-four business leaders, all of whom strongly affirmed the importance of organisational culture. Our review contributes to the existing literature by integrating research from these different disciplines and by offering an integrative and holistic review that combines both the insights from academic research and experience based on practice along four dimensions: (1) how the construct is to be defined; (2) what underlies its importance; (3) how it can be measured; and (4) how it can be operationalised, that is, established within an organisation.

Section 2 begins by providing an overview of how the construct of corporate culture has been defined in the academic literatures. We motivate our decision to include research on organisational climate within a survey of culture, arguing that arguments to treat the two constructs as conceptually distinct are not compelling. We then propose a framework that aims to capture five key dimensions of the construct of corporate culture.

Section 3 focuses on the importance of culture. Whereas culture often tends to be seen as the 'soft' side of business, the literature shows that culture is significant with respect to the implementation of strategy, business performance, and how a corporation deals with the socio-political context within which it is situated. In addition to reviewing the literature, we draw on learning from cases of organisational success and failure that has emphasised the importance of culture and its inherent connection to other domains such as politics.

Having examined different ways in which culture has been defined and why culture, so understood, is important for an organisation, Section 4 considers how culture can be measured. We survey a range of different approaches that have been developed to measure the construct, observing that there is no consistent use of any one measure to-date.

Section 5 explores different ways in which a certain culture can be operationalised (for example, in physical space) and changed (for example, to increase diversity).

Given our focus on an integrative and applied review, this section draws on best-practice examples of culture implementation and change from different industries (technology, engineering, and consumer goods).

Section 6 concludes our discussion, highlighting some emerging gaps that merit further research, such as the connection between the social purpose of the corporation and culture, as well as suggestions to better support the alleged connection between culture and ethical outcomes.

# 2 THE CONCEPTUAL FOUNDATIONS OF THE CONCEPT OF 'ORGANISATIONAL CULTURE'

#### 2.1 Culture vs. climate?

In the late 1970s and early 1980s, organisational culture research emerged from the disciplines of anthropology and sociology, which are concerned with how important values arise and develop in social groups of differing sizes. Organisational culture research uses qualitative case methods based on observations of behavioural patterns of group members at different points of time and a focus on how values within the group are transmitted through myths and stories. Inspired by this way of studying collectives, the construct of organisational culture was first introduced to the study of organisations through seminal work by Pettigrew (1979), as well as work by Deal and Kennedy (1982) and Peters and Waterman (1982).

The development of the field of organisational culture research was preceded by what is known as organisational *climate* research. In contrast to the anthropological and sociological influences of the organisational culture literature, climate research has its roots in the Gestalt psychology of Lewin *et al.* (1939), who argued that the psychological life space that people inhabit is mainly a product of the social and behavioural attitudes of people to leadership practices, which they consequently termed 'social climate'. In the 1960s, Lewinian methodology was introduced into the study of organisations to gain an insight into the social and behavioural perception of an organisation by its members.

Up until the beginning of the 21st century, organisational culture and climate research operated largely orthogonally. However, more recent research has begun to acknowledge that the distinction between culture and climate might have been exaggerated or even misguided and now either treats the two constructs as identical or develops integrative approaches that combine both fields of research and methodologies (Denison 1996, Ostroff *et al.* 2003, Schneider & Reichers 1990, Treviño *et al.* 1998, 2006, Zohar & Hofmann 2012).

In line with this trend, we shall conceive of the two constructs as largely congruent throughout this review. We therefore survey, in addition to culture models, measurement frameworks from the 'traditional' climate camp as well as hybrid approaches that draw on both fields (Treviño *et al.* 1998; Schneider *et al.* 2011). Opting for this integrative strategy aligns with our interview results with business leaders, none of whom considered the distinction between culture and climate to be of importance in practice (British Academy 2018).

#### 2.2 How should culture be defined?

Our survey of conceptualisations of culture suggests that there is a considerable variety in definitions of culture, and very little convergence towards a single definition. Cameron and Ettington (1988) report eighteen different definitions in the literature from 1982 to 1986 alone while Verbeke *et al.* (1998) note fifty-four different definitions in the period from 1960 to 1993.

As Pettigrew observes, one explanation for this might be that culture is a very rich construct that can subsume or encompass other concepts:

Part of the problem with culture is that it is not just a concept but the source of a family of concepts ... and it is not just a family of concepts but a frame reference or root metaphor for organisational analysis (Pettigrew 1990: 414–15).

Our review of the literature confirms Pettigrew's hypothesis: while many definitions of culture centred around common themes—which we will try to capture more formally in the next section—definitions vary considerably on closer inspection.

Some accounts define culture rather broadly, such as:

• Groysberg *et al.* (2018: 4) who consider culture to simply be 'the tacit social order of an organisation'.

Other accounts emphasise that culture primarily concerns entrenched beliefs. Examples here are:

- Louis (1980: 227) who defines culture as 'a set of common understandings for organizing actions and language and other symbolic vehicles for expressing common understandings'; and
- Schwartz & Davis (1981: 33) who define culture as 'a pattern of beliefs and expectations shared by the organization's members'.

Other definitions also consider more implicit elements as constitutive of culture, such as behavioural norms or symbols:

- Turnstall (1983: 1) understands the construct as 'a general constellation of beliefs, mores, value systems, behavioural norms and ways of doing business that are unique to each corporation';
- Alvesson (1993: 2–3) defines culture as 'a shared and learned world of experiences, meanings, values, and understandings which inform people which are expressed, reproduced, and communicated partly in symbolic form'.

In this context, it is worth highlighting the extent to which definitions of climate capture similar notions to the ones invoked above. For example:

- Ashkanasy *et al.* (2000: 8) define climate as a term 'being used to describe configurations of attitudes and perceptions by organization members that, in combination, reflect a substantial part of the context of which they are a part and within which they work'; and
- Michela and Burke (2000: 234) write that 'traditionally defined, organisational climate involves people's perceptions and experiences of the workplace in terms of warmth, trust, dynamism, ambiguity, and other affect-laden dimensions'.

One of the most encompassing and canonical definitions is given by:

• Schein (2016: 18), who defines culture as 'a pattern of shared basic assumptions learned by [an organisation] as it solved its problems of external adaptation and internal integration, which has worked well enough to be considered valued, and therefore to be taught to new members as the correct way to perceive, think and feel in relation to those problems.'

Lastly, some economic literature examines culture from a perspective that is internal to an organisation and, specifically, aims to understand how the presence of culture can improve the efficiency of internal communications and decisions (see also Arrow 1974). In context:

• Crémer (1993: 354) proposes that culture is 'the part of the stock of knowledge that is shared by a substantial portion of the employees of the firm, but not the general population from which they are drawn.'

# 2.3 An integrated framework of culture

We observed in the previous section that, although there exists a great variety of subtly different definitions of what organisational culture is, most definitions centre around similar ideas. To this end, we shall synthesise our survey results to provide a frame-

work which captures some central features of the construct of culture in a more concise way than existing proposals.<sup>1</sup>

The articles that we surveyed as part of this review suggest that culture can be defined as sharing (at least) the following five key features:

#### Organisational culture as a social phenomenon

Organisational culture is a social construct that only manifests in a collective: that is, organisations with a certain number of members.

#### Organisational culture as concerning values

At the most general level, accounts of organisational culture define culture as having to do with values, understood very broadly and as manifesting and being shaped in various ways.

We propose to distinguish this feature into two sub-categories: the mental and the physical. The mental dimension concerns either consciously or tacitly shared beliefs, attitudes, or other perceptions such as narratives. The physical dimension concerns, depending on the account, observable artefacts such as symbols, organisational language, and practices. These features are physical in the sense that they are directly observable in the behaviour and artefacts created by people. For example, some frameworks maintain that the physical manifestation of organisational culture can be observed in the form of formal practices, such as job descriptions, policies, and information practices, in the way stories about organisation are told, rituals such as ceremonies and product unveilings, and, more generally, in the way decisions are made.

#### Organisational culture as related to action

Accounts of organisational culture all see culture as necessarily related to action. This relation can be understood broadly and in different ways: in terms of culture facilitating or hindering certain kinds of (more efficient) action (as we shall see in Section 3), or in terms of culture being influenced by agency itself (as noted in the previous paragraph).

However, many accounts endorse a more stringent connection than this, linking culture directly to *ethical* action as opposed to broader forms of agency. These conceptions see culture as related to, for example, promoting behaviour that conforms with an organisation's norms, rules, and practices.

<sup>&</sup>lt;sup>1</sup>Some other examples of such classifications are Alvesson (2002), Martin (2002), Ott (1989), Schein (1991, 2010), Schein (2016), and Trice & Beyer (1993).

#### Organisational culture as scalar

Accounts of organisational culture mostly agree that the notion is scalar: viz. that an organisation's culture can be developed to differing degrees. An organisation's culture can be more fragmented (weak), as might be the case in some newly formed organisations or organisations that have recently undergone radical changes, or fully developed (strong), as in some long-established organisations or well-managed organisations. For example, one might think that the strength of an organisation's culture relates to how effectively it can shape conduct. (See, for example, the idea that climate strength is defined as the extent to which members of a work unit or an organisation agree in their perceptions of the workplace climate (Lindell & Brandt 2000).)

# Organisational culture as multi-layered

Some frameworks of organisational culture adopt what we might call an intraorganisational *pluralist* as opposed to *monist* view of culture. According to pluralist conceptions, there does not necessarily exist only a single form of culture within any organisation (Schein 2016). Different sub-cultures/distinct cultures can exist at different organisational levels: that is, team-level, and board-level. (Scholars have found a range from three (Wimbush *et al.* 1997) to nine (Peterson 2002).)

#### 3 THE IMPORTANCE OF CULTURE

With a clearer idea of how to conceptualsie the construct of corporate culture, let us now consider why different aspects of culture are important for an organisation. For a long time, culture was considered to be of lesser importance for organisational performance than, for example, strategy. However, more recent literature suggests a greater importance of culture. The following subsection summarises arguments highlighting the importance of culture with respect to three main areas—strategy implementation, business performance, and politics.

First of all, it should be noted that the importance of corporate culture (and elements thereof) can differ according to industry. Contingency theorists (e.g., Lawrence & Lorsch 1967, Thompson 1967) highlight that not all the elements of culture which have been pointed out above, are equally important. As straightforward as this might seem (that is, that in manufacturing the quality culture is more salient whereas in a sales organisation, the focus will be on a culture that encourages good relationships with customers), it is nonetheless important to bear in mind for the following discussion (Detert *et al.* 2000). It is also important to remember that we are relying on an integrated understanding of culture, which means that the importance of the respective

five elements outlined in the previous section differs in each of the respective importance dimensions below.

# 3.1 'Culture eats strategy for breakfast'2

Most recent contributions to this field emphasise the relationship between culture and strategy and have shown that culture can function as the facilitator of strategy. Whereas strategy 'provides clarity and focus for collective action and decision making' (Groysberg *et al.* 2018: 4), culture facilitates (or hinders) the successful implementation of strategy. As the last section of this paper will further elaborate it, this relationship is particularly strongly pronounced with respect to the implementation of technological change. According to a study by the McKinsey Global Institute (MGI), culture is the most significant hindrance to digital effectiveness. The lack of appropriate cultural change is perceived to be even more important than, for example, the lack of IT infrastructure.

MGI's research suggests that it is crucial for corporations to intentionally change their culture in order to be better able to deal with the changes effected by digitisation rather than waiting for the culture change to happen by itself. Goran *et al.* (2017) show a clear correlation between negative business performance and cultural obstacles due to a slowly changing corporate culture. It is suggested that an early adoption of and investment in cultural change in response to digitisation can make a significant difference and allow corporations to be more responsive to customer demands, less risk averse, and to ensure greater interaction across different business units.

A similar conclusion is reached by Groysberg et al., who emphasise that corporate culture, here understood with a particular emphasis on the 'culture as a social phenomenon' and 'culture as concerning values'—that is, as the implicit social order of an organisation which shapes its employees' values and attitudes—is essential in maintaining an organisation's viability and effectiveness (Groysberg et al. 2018). As '[C]ulture expresses goals through values and beliefs and guides activity through shared assumptions and group norms' (3), it is therefore essential in facilitating the deliverance of strategy. However, the literature also emphasises that this insight regarding the importance of culture is not (yet) shared by all leaders. It has been observed that, despite its importance, the propagation of culture is often seen to be of lower priority and the human resources department, to which culture development if often delegated, sometimes lacks power. One of the reasons for this is that culture's elusive nature (in addition to the conceptual unclarity seen in the section above), makes it

<sup>&</sup>lt;sup>2</sup> Although there are debates about the true origin of this quote, it has commonly been attributed to the management consultant Peter Drucker.

harder for organisations to derive specific means and measures which would facilitate its measurement and active management.

Despite this, culture is claimed by many to be crucial: 'When properly aligned with personal values, drives, and needs, culture can unleash tremendous amounts of energy toward a shared purpose and foster an organisation's capacity to thrive' (Groysberg *et al.* 2018: 4). This claim is based not only on Groysberg *et al.*'s review of the academic literature but also on their practical experience. Importantly, they emphasise that culture's positive effect in delivering strategy and hence leading to better business performance relies on the alignment of culture and strategy (Groysberg *et al.* 2018: 8). The thought seems to be that, if culture and strategy are coherent, by having internalised the corporation's values, employees will act and decide such that they work towards delivering the respective strategy.

On the other hand, culture's importance can also be deduced from negative examples. The above-mentioned MGI study also emphasises that a 'false' corporate culture, understood as misaligned with the corporation's strategy (and purpose), can also inhibit changes. This is particularly pertinent with respect to adapting to digitisation. For instance, in a software company which aims at developing new software products, it is essential that developers have the freedom to exchange their ideas as well as test and try them for proof of concept. Here, examples of the ineffective or damaging cultures are: thinking in silos (rather than all parts of the company being organised as a network where information travels easily); excessive risk aversion; an overall non-digital culture (lack of experience with digital tools). Along these lines, the Salz Review of the banking sector (Salz 2013) also finds that the culture of an organisation, here with a special emphasis on trust, can 'defeat its formal governance' (FSB 2017: 5).

#### 3.2 Culture and business performance

If culture is so strongly linked to strategy, it will not come as a surprise that it also is intricately connected to business performance. Heskett, for example, highlights that culture may 'account for 20–30% of the differential in corporate performance when compared with 'culturally unremarkable' competitors' (Heskett, as quoted in Coleman 2013: 2).

Along these lines, it has also been pointed out that culture can facilitate financial stability. The authors of the FSB (Financial Stability Board) report, for example, emphasise that culture (here understood in the context of corporate governance) 'is critical to financial stability' (FSB 2017: 1) and hence to performance. Other studies, that have tried to quantify the relationship between culture and business performance confirm that there is a positive relationship. Boyce *et al.* (2015) have collected data

from ninety-five automobile dealers over six years in order to find out whether culture influences performance or whether the direction of the relationship is the other way around. Their results show 'that culture "comes first" '(Boyce *et al.* 2015).

The relationship between culture and business performance has also been highlighted in examples of poor performance. The Salz Review, which is a review of culture in Barclays Bank after the LIBOR rigging scandal (Salz 2013), finds that, '[i]n many instances of misconduct, there is evidence that the norms and expectations that most strongly influence behaviour within financial institutions can be very different from the institutions' stated values and policies' (FSB 2017: 4). The Salz Review emphasises that the norms and values encapsulated in a corporation's culture influence business performance—not only in the narrow sense of economic performance, but also in the sense of conduct: that is, compliance with the law.

Whereas these studies emphasise the relationship between culture and performance from an internal perspective, the link between culture and performance from an external perspective has also been made. Kreps (1986), for example, highlights that a good reputation can be seen as an asset that can attract investments. Reputation, in turn, is closely related to a firm's culture, as the corporate culture influences how (positively or negatively) employees speak about their firm. A prominent recent example of this is the case of Uber's founder and then-CEO, Travis Kalanick, being made to leave after a senior engineer publicly wrote about the culture of sexism at Uber (O'Brien 2018). It seems clear to analysts that its ruthless culture has been the driver for this and other scandals (such as covering of data hacks) and, respectively, for its negative reputation. The board's action to let go of the CEO-founder is an indication of the fear of negative impacts of the culture he instilled on the company's reputation and business performance. Thus, the example of Uber supports Kreps' theory in the negative and shows the influence of corporate culture on reputation and the possible link to performance from an investor's perspective.

#### 3.3 Culture and political influence

In addition to culture being discussed as important in relation to business performance and strategy, culture has come to be seen as important in relation to politics. For instance, it has been pointed out that in non-Western, non-capitalist contexts, such as China, the manipulation of corporate culture can be a tool that allows government to further its (socialist) agenda (in non-capitalist countries). Hawes (2008), for example, holds that, '[b]uilding a progressive corporate culture is a significant factor in strengthening the [Communist] Party's hold on power, in forcefully developing a progressive socialist culture, and in building a harmonious socialist society.' (Chinese Supervision and Administration Commission of the State Council (SASAC) Guiding Opinion

Article (2005) as cited in Hawes 2008: 40). By a similar token, it has been pointed out that a particular political culture can have a strong influence on a corporation's culture. It has been suggested that this has been a contributing aspect to the situation that TeliaSonera found itself in: that is, having to deal with how the political culture in its Eastern European operations influenced its business culture, which differed from the culture in its headquarters in Sweden (FOTC virtual roundtable discussion, April 2018).

To conclude, it should be noted that the literature emphasises the importance corporate culture can have in furthering political agendas or in adopting a certain political culture. However, what has not been sufficiently established in the literature is whether this is more than a correlation and what is the direction of influence between politics and corporate culture.

Thus, a survey of the literature as well as a consideration of case examples indicates that corporate culture is important, amongst other factors, in the areas of strategy implementation, business performance and with respect to how a corporation deals with the given politics of the market where it does business.

#### 4 THE MEASUREMENT OF CULTURE

# 4.1 Methodologies

Frameworks that aim to measure the construct of culture itself rely on quantitative or qualitative measures, or a combination of both. Qualitative measures consist of controlled interviews with an organisation's leadership and employees as well as employee, customer, and other stakeholder surveys and focus groups (Salz 2013, Savitz 2013). While these surveys and interviews have traditionally formed the core of culture research, these qualitative methods are increasingly combined with quantitative measures (Mayer 2014: 273). Most recently, some have employed meta-analyses of employee online behaviour, such as the frequency and idiomatic patterns of e-mail exchanges, as contributing inputs to measure an organisation's culture (British Academy 2018).

#### **4.2** Measurement frameworks

To provide a top-level summary of our results, we have captured the commonalities of these frameworks in Table 1, surveying whether frameworks employ a strictly ethical conceptualisation of culture or a mixed one that measures other elements, the number of attributes that the framework tracks, and how many dimensions it has. One observation to make here is that the popularity of their application does not seem to correlate with the complexity of frameworks and number of attributes they track.

Commonalities of frameworks	Ethical vs. mixed-ethical	Number of attributes tracked	Single vs. multidimensional
OCAI	mixed-ethical	4	multidimensional
OCM	mixed-ethical	3	multidimensional
ECQ	ethical	2	multidimensional
CEV	ethical	5	unidimensional
OCS	mixed-ethical	4	multidimensional
OCP	mixed-ethical	5–7	multidimensional
ECI	ethical	10	multidimensional
CEVM	ethical	7	multidimensional

Table 1. Comparison of widely used measurement frameworks.

In what follows we provide a more elaborate survey of the measures compared in the above table, to give the reader a flavour of how extant approaches attempt to measure the construct of culture. These measurement frameworks either rely on extant definitions of culture similar to or adopting one of the definitions that we surveyed in Section 2, or they propose conceptualisations of their own that can be captured by our proposed integrative definition.

# Organisational Culture Assessment Instrument (OCAI)

This influential framework is based on the so-called competing values framework (Quinn & Rohrbaugh 1983) and consists of six questions. These are labelled dominant characteristics, organisational leadership, management of employees, organisation glue, strategic emphases, and criteria for success. Each question contains alternatives that correspond to one of the four organisational values of the Competing Values Framework: clan (commitment to employees, participation and teamwork), adhocracy (dynamic and entrepreneur organisations, cutting-edge output, innovation), hierarchy (formalised and structured organisations, smooth functioning, stability), and market (competitive organisations, increasing market share, productivity). There is evidence that the framework is correlated with important organisational outcomes, such as employee job satisfaction and commitment, customer satisfaction, and market value (Gregory *et al.* 2009, Hartnell *et al.* 2011).

# Organisational Culture Model (OCM)

The OCM model aims to provide a dynamic account of organisational culture and consists of three levels: (1) artefacts and symbols, (2) espoused values, and (3) basic underlying assumptions (also known as the Iceberg Theory given that, according to the Scheins' framework, most of the constituent elements of culture are unlikely to be visible). Artefacts concern the surface of an organisation and concern visible elements

of an organisation. Espoused values concern an organisation's standards, values, and rules of conduct. Thirdly, basic underlying assumptions are deeply embedded in organisational culture and experienced as self-evident and unconscious behaviour.

# Ethical Climate Questionnaire (ECQ)

The ECQ model is another popular model (Victor & Cullen 1987, 1988). Victor and Cullen define climate as 'the shared perception of what is correct behaviour, and how ethical situations should be handled in an organisation' (Victor & Cullen 1987: 51). The framework has two dimensions to highlight different types of ethical climates. The first is the ethical criterion, which aims to cover three main schools of ethical theory: (a) egoism (maximising of self-interest), (b) benevolence (maximising joint interest), and (c) deontology or principle (adhering to a principle) (Fritzsche & Becker 1984, Williams 1985). The second dimension is called the *locus of analysis* and focuses on one of three potential referents for one's actions: the self (oneself), the local (one's organisation or subunit), and the cosmopolitan (the environment external to the organisation).

#### Corporate Ethics Values (CEV)

The CEV model's main idea is that an organisation's culture concerns its ethical values. Hunt *et al.* define (1989: 80) corporate ethical values as 'values that help establish and maintain the standards that delineate the "right" things to do and the things "worth doing" and develop a five-item measure to assess the ethical values so construed. Several subsequent culture assessments have relied on this operationalisation (Burnaz *et al.* 2010, Douglas *et al.* 2001, Rubin *et al.* 2010, Sweeney *et al.* 2010).

#### Organisational Culture Survey (OCS)

Denison (1990) and Denison & Neale's (2000) framework uses the same underlying dimensions of the CVF introduced previously, combining the ideas of flexibility versus stability and internal versus external orientation to create a four-cell matrix (clan, adhocracy, hierarchy, market). The OCS is a 60-item survey that consists of four dimensions, which each split into three further sub-dimensions: involvement, consistency, adaptability, and mission. These correspond, inter alia, to the extent to which employees can affect decisions, the extent to which an organisation's beliefs and values align with its policies in practice, the extent to which the organisation can adapt to its environment and customers, and the extent to which the organisation has a clear vision and purpose (Denison 1990, Denison *et al.* 2012, Kotra *et al.* 2012).

#### *Organisation Culture Profile (OCP)*

The OCP is another common cultural assessment tool. It is a 54-item survey that asks participants to sort values or attributes from most to least characteristic of their organisation. Relevant dimensions captured by the OCP vary according to study. For example, O'Reilly *et al.* (1991) proposed eight dimensions captured by the OCP: innovation, stability, respect for people, outcome, orientation, attention to detail, team orientation, and aggressiveness; while Chatman and Jehn (1994) modified some of these and Lee and Yu (2004) only confirmed five.

#### Ethical Culture Index (ECI)

The ECI model seeks to integrate previous literature on culture and climate and defines culture as a 'multidimensional interplay among various "formal" and "informal" systems of behavioural control that are capable of promoting either ethical or unethical behaviour' (Treviño *et al.* 1998: 451). Examples of formal parts of this system include policies, leadership, authority structures, reward systems, and training programmes; examples of informal systems focus on the behaviour of others with respect to ethical norms. The model integrates a 21-item measure based on this definition (subsequently used in several articles: e.g. Key, 1999, Schaubroeck *et al.* 2012, Shi & Chen 2006, Tsai & Shih 2005, Zhang *et al.* 2009a, 2009b) with a 26-item ethical climate questionnaire developed by Victor and Cullen (1987, 1988), resulting in a ten-dimensional measure which covers: ethical environment, employee-focused climate, community-focused climate, obedience to authority, code implementation, self-interest climate, efficiency climate, rules and procedures climate, personal ethics climate, and law and professional codes climate.

#### Corporate Ethics Virtues Model (CEVM)

The CEV model draws on Solomon's (1992, 1999, 2000, 2004) virtue-based theory of business ethics. Kaptein suggests that virtuous companies can develop a culture that stimulates ethical behaviour and prevents unethical behaviour, and defines corporate ethical virtues as 'the organisational conditions for ethical conduct; they reflect the capacity of an organisation to stimulate ethical conduct of employees' (1998: 924). The CEV model assesses seven key organisational virtues that comprise an organisation's culture: clarity, congruency, feasibility, supportability, transparency, discussability, and sanctionability (Kaptein 1998). The CEV keeps being continuously developed (Kaptein 2011a, 2011b) and several articles have employed the model (Huhtala *et al.* 2011, Webb 2012).

#### 4.3 Consistency of employed measures

Most studies use variations of the measurements of culture introduced above that either modify a single measure for a study, combine it with other frameworks, or devise an entirely new measure (for example, some adopt the full 36-item measure, while other studies employ 26- or 16-item versions of the ECQ). For example, work by Pierce and Sweeney (2010) employs five items from Hunt *et al.*'s (1989) approach and combines it with six items from Treviño *et al.* (1998). Work by Koh and Boo (2001) uses a combination of items from Hunt *et al.*, Treviño *et al.*, and Victor and Cullen (1987, 1988). Other studies develop their own conceptualisations, but do not make them explicit in their research (Wittayapoom & Limsuwan 2012). As Mayer thus (2014: 433) observes, 'there is little consistency in the use of measures or manipulations of ethical culture in the literature. Although more than 30 articles have examined ethical culture, the most any one measure or manipulation of ethical culture has been used is six times.'

We note that the rationale for why a given measure of organisational culture has been chosen for a study or why a chosen measures' items have been selected or modified is often not articulated in studies. For example, Webb (2012) merely notes that the measure by Kaptein (2008) proved fruitful for the purpose of the study without specifying further the merits of the chosen approach. Other studies fare better, arguing that some of the adopted items—in this case the focus on law and code and rule dimensions from Victor and Cullen's (1988) typology—are more positively associated with work outcomes at which the research is aimed.

It is worth highlighting that the lack of provision of a rationale for the employment of a given measure is problematic, insofar as not all of them are equally suitable. Rather, the measurement chosen is likely to depend on certain variables, such as the size of an organisation (Halaj 2017: 626). One neglected issue for the measurement literature of organisational culture is consequently to consider the criteria that make the application of a given framework more or less suitable.

#### 5 THE OPERATIONALISATION OF CULTURE

With a better understanding of how corporate culture can be measured, we can now turn to discussing the extent to which culture can be a supporting factor when operationalising organisational changes or making an organisation more attractive as a place to work. Independent of the aim, in order for culture to be more than empty words written on the organisation's wall, culture needs to be embedded in the organisation's practices (Coleman 2013). McKinsey, for example, brands itself as daring to

tell its clients the 'uncomfortable' but 'real truth'. Likewise, its employees are encouraged to dissent if they think that a superior's proposed solution will have negative results (Coleman 2013: 3).

# 5.1 Corporate culture and innovation

One of the key examples of operationalising a corporate culture that fosters innovation has been termed 'innovation parenting' (Ishak 2017). This term is based on a comparison between how parents ensure their children learn the values that are important to the family whilst giving them enough space to develop their own personalities.

'Innovation parenting' refers to a number of different aspects, starting with making sure that employees internalise the values of the company. First of all, this entails that the innovation leaders (for example, chief technologists) live the core values and strategic priorities of the organisation, lead by example, and allow their colleagues enough freedom to explore new business opportunities or different technologies. In other words, corporate culture needs to be *internalised* to allow for successful innovation. Ishak argues that it is crucial to develop a culture of trust in order to give employees sufficient freedom to explore creative ways of problem-solving (Ishak 2017).

For leadership teams, this also means ensuring flat hierarchies and avoiding micromanagement. Arguably, this is what Silicon Valley start-ups have realised and is part of their success. A prominent example of this is Facebook's famous 'Move fast, and break things' motto. To achieve this, Facebook is organised along flat hierarchies and small teams that are actually able to move fast—in terms of team size but also in terms of permissions to take decisions. As Facebook itself states, it is defined by its 'hacker culture—an environment that rewards creative problem solving and rapid decision making' (Facebook 2018). Of course, the ongoing 'Cambridge Analytica scandal' raises questions with respect to what this culture does to the ethical behaviour of the company, but it does appear to further innovation (Feloni 2016).

# 5.2 Culture change and employee's demands

Research shows that workers prefer flexible and independent work. Companies wanting to respond to workers' preferences should change their culture and can use the latest technology to support this. An example of this are apps that allow employees on storefronts to see where they might be needed more at peak times. Not only does this a quicker response to customer needs but it also gives employees a higher degree of autonomy at their workplace. Autonomy in turn is a value that Millennials

particularly demand (FOTC one-day conference, British Academy, May 2018). Importantly, a corporation's culture is crucial in ensuring the quality of work while work is organised in a more flexible and independent way.

The literature also highlights the extent to which corporate culture influences the (mis-)conduct of their employees. For example, culture plays a necessary role in the workplace. Since '[v]alues drive everyday behaviour, helping to define what is normal and acceptable, explaining how things ought to be (for example, staff ought to put customers first). Values provide a framework through which the natural and often difficult conflicts that arise in people's day-to-day work can be resolved' (Salz 20013: 79–80). The FSB report also finds that culture is an important factor with respect to misconduct or industry hazards: 'In some cases the culture was limited to the organisation, but in others it was the culture of the industry as a whole' (FSB 2017: 65).

Corporate culture is also linked to having an impact on individuals' behaviour (outside the workplace). For example, Schlicht (2004) who analyses whether individuals in 'capitalism' tend to be, as is often assumed, selfish and competitive and finds that, 'In cases where motivation, involvement, and commitment are important, self-ishness may be outcompeted by generosity' (241). Along similar lines, Williamson *et al.* (2009) find that there is a link between culture (responded to by different kinds of rewards) and employees' behaviour (in this case: commitment). The example of the Chinese government relying on corporate culture to form loyal citizens also points to this. Here, as Hawes argues, corporate culture helps corporations to 'produce' employees who are well educated, cultivated, and highly motivated as well as loyal citizens (Hawes 2008: 42).

# 5.3 Corporate culture and space

An organisation's building or the way the offices are designed can be a significant contributing factor in operationalising culture. There are a number of examples that illustrate this relationship. Commonly cited is Pixar's open atrium, which facilitates employees running into each other coincidentally and chatting about new ideas and problems they are currently facing (Coleman 2013). Google is known for creating office environments that resemble living rooms, jungles, or libraries and thereby hoping to inspire their employees' creativity. Although the literature lacks large-scale empirical examples establishing more than a correlation between innovative office spaces and innovative business ideas, the literature seems to be rather confident that these kinds of relationships (for example, between open office spaces and increased collaboration) exist (Coleman 2013).

Culture can also be implemented through the choice of office locations. In Europe, major concentrations of start-ups exist in London, Berlin, and Lisbon. Although

some authors recognise the initial reasons for this being linked to inexpensive office space, it is widely held that many organisations are now choosing these cities as employees find it attractive to live there. Companies based in Berlin, and preferably having chosen to revamp an old industrial complex as their office space, are considered to be 'avant-garde' (examples are Zalando, RocketInternet).

# 5.4 Corporate culture change and (gender) diversity

'Empowerment and skill-building, not shaming and blaming, are key to engaging men as advocates for change.'

Prime (2013)

The attention on issues surrounding diversity has increased significantly. Two main reasons seem to be the key drivers of the increased attention on diversity. The first is that research has shown that diverse teams are more productive.<sup>3</sup> The second one is egalitarian: focussing on the injustice of excluding certain gender or ethnic groups from private corporations. It is not the aim of this paper to assess whether or not the reasons for focussing on increasing diversity in corporations are justified.<sup>4</sup> What we focus on instead is the research which has been conducted with respect to the question to what extent corporate culture has an influence on diversity.

#### 5.5 Single-gender dominated culture

Examples of male-dominated business culture are plentiful. A study by the head-hunting firm Harvey Nash together with Inspire has found that '[m]ale-dominated corporate culture is the biggest barrier for women reaching board level' (Newcombe 2013): 25 per cent of the respondents (600 directors, CEOs, and senior executives) report the belief that male-dominated culture is the 'single biggest barrier to progression' (Newcombe 2013). It is not uncommon that a predominantly male corporate culture will lead women to leave their employer, with the result that, if women leave their corporate careers early, 'organisations risk disconnecting with their customers, weakening their competitive edge and missing out on further opportunities. This isn't a talent issue, it's simply a bottom-line business issue' (A. Bailey, as cited in Newcombe 2013).

<sup>&</sup>lt;sup>3</sup> According to Desvaux *et al.* (2010) 'companies with top quartile representation of women in executive committees perform significantly better than companies with no women at the top.' On average, they have 47 per cent more return on equity and 55 per cent more average earnings before taxes (Devillard *et al.* 2013).

<sup>&</sup>lt;sup>4</sup> For further information regarding the importance of female representation on boards, see Higgs (2003).

How prevalent can such a predominantly male (or female) culture be, one might wonder. Cohen (2013) finds that there are areas in which gendered segregation is the norm. In the United States, 26 per cent of workers are in professions which are dominated by a single sex (90 per cent single sex). Prominent examples include truck drivers and nurses. Among the 69 per cent of workers who work in 'merely very-segregated' occupations, they are 'at least two-thirds single sex' (Cohen 2013). This includes, primary teachers, for example. Cohen takes these to highlight how common sexual harassment can be where the culture is male dominated: She concludes that 'just like peer culture can prevail over parents' grownup interventions when it comes to socialising adolescents, workplace culture spills over into family life, as men in male-dominated jobs (such as police officers) or female-dominated jobs (where their masculinity is threatened) [can] perpetrate violence at home' (Cohen 2013).

#### 5.6 Mainstream-identity dominated culture

The negative effects of an exclusive corporate culture are, however, not only linked to gender. Prime (2013), for example, emphasises the negative effects of a predominantly white male culture. She argues that '[t]aking advantage of workplace diversity is one of the critical challenges of leadership. Organisations with a track record of developing leaders from a particular background are likely to be suffering from diversity, inclusion, and leadership problems at the same time.' The question is, where does a 'white-male-dominated culture' begin and how can you change it. The example of Rockwell Automation (a global engineering company) trying to diversify its North American sales division provides a good example. Instead of just having the aim of 'diversification' tagged to its recruiting strategy, Rockwell focussed on changing its (white-male-dominated) culture first. This involved, first and foremost, understanding what makes the experiences of white males different from those of other identity groups. This involved a series of workshops that would allow Rockwell's employees to reflect on the advantages their identity would bring them and how their experiences at work were different from those of people from other identity groups. A study that was conducted after these workshops showed that these exercises did 'have a transformative effect, shifting both the mindsets and behaviors of participants' (Foust-Cummings et al. 2012).

Among other things, this example shows well the extent to which *awareness* and shifting the *mindset* can be crucial first steps to changing the culture. As Prime puts it, 'This [was] a critical lesson for other organisations: rather than feeling responsible for group-based inequities that they did not create, white male managers should feel empowered and equipped to lead the creation of an inclusive workplace' (Prime 2013).

#### 5. 7 Leadership and culture change

Before we close, let us consider the role of leadership in culture change. Here we want to pay particular attention to the ways in which leadership has been seen to relate to organisational culture change and to what extent culture can be managed.

The literature on leadership culture change can be divided into two extreme camps: the view according to which leaders have little to no influence on culture and the view according to which they have strong influence, with a more moderate position falling between these two. Examples of incompatibilist positions are those of Meek (1988), Alvesson and Berg (1992), and Martin and Siehl (1983: 53). The latter see the influence that managers can have on culture as only very limited: 'It may be that cultures cannot be straightforwardly created or managed by individuals. Instead, cultures may simply exist and managers may capitalize on culture effects they perceive as positive or minimize those perceived as negative.' Compatibilist positions are proposed by Tichy (1982), Turnstall (1983), Cameron and Quinn (2011), and Deal and Kennedy (1982), who break down culture change into a series of steps.

In addition to the above discussion regarding the relationship between culture and innovation, it has also been proposed that leaders have a significant impact on creating an innovation culture. Closely related to this are well-known examples of companies where the founder of the company has a significant impact on corporate culture. Mark Zuckerberg for Facebook or Steve Jobs for Apple come to mind. With respect to Facebook, this influence is particularly visible. As a company that was (allegedly) started in Zuckerberg's dorm room, it is among the few top US corporations where the CEO only wears a suit when he hast to testify in front of Congress. Appearance aside, the influence of Zuckerberg's ambition and technology-driven mindset defines Facebook's culture as whole (British Academy 2018).

More moderate positions acknowledge that the extent to which leaders can influence and manage organisational culture depends on a multitude of factors, such as the stage of an organisation's life cycle, whether it is in its early or more mature stage; the level of the organisation (subunit or entire organisation); the presence of a need for change (for example, due to a crisis); and the ability of the relevant leaders (Louis 1985; see also Alvesson & Sveningsson 2008). For example, Lundberg (1985) proposes a range of conditions that make change in an organisation's culture feasible for its leadership, where whether these conditions obtain determines the likelihood that the change will be successful. He proposes four conditions: external enabling conditions, which concern the existence of threats in the external environment; internal enabling conditions, which relate to the intra-organisational capabilities to make culture change successful; precipitating pressures, which concern performance demands, or crises (real or perceived); and triggering events, which are understood as

the most proximal conditions for culture change, involving large changes to the organisation's environment, such as radical changes in the management team.

This section has shown that corporate culture can be operationalised and changed in a number of different ways. The areas that the current literature highlights are the extent to which a certain corporate culture can further (hinder) innovation. We have also discussed how corporate culture can be used to capture employees' changing demands and needs and that the implementation of culture is not restricted to workshops and value statements, but can be as concrete as designing the workspace in a particular way. Lastly, it has been shown that culture is an important element in increasing diversity and that, on the whole, leadership is seen to be instrumental to implementing (or changing) culture.

#### 6 CONCLUSION AND OUTLOOK FOR FUTURE RESEARCH

#### **6.1** Future research on culture

In the context of the future of the corporation, this paper has reviewed recent contributions on corporate culture. We have started by organising the various ways in which culture is understood in the literature (Section 2), why it is important (Section 3), how it can be measured (Section 4), and, lastly, how culture can be implemented and changed (Section 5). We proposed that the various definitions of culture display five key features: culture is understood as a social, scalar, multi-layered phenomenon that is concerned with values and related to action. Section 3 highlighted that culture is significant with respect to the implementation of strategy, business performance, and to how a corporation deals with the socio-political context within which it is situated. With these insights in mind, Section 4 provided an overview of the different measurement methods to capture the (potential) effectiveness of culture. The final section discussed different ways in which culture can be operationalised.

Based on our survey of the culture literature, the following points of interest for further research for each of our dimensions can be identified:

- A clear conceptual definition of culture that encompasses the five features that
  we highlighted and explains how, if at all, culture relates to the conceptual
  construct of organisational climate.
- Further evidence on the (causal) relationship between areas in which culture is considered to be important (especially with respect to culture and politics).
- There is a need for a closer examination of the conditions that influence the appropriateness of different measurement frameworks of the construct of culture.

A coherent study of best (and worst) practice examples of the operationalisation
of culture. Currently, the literature is (mostly) based on anecdotal evidence, with
different measures for the success of the operationalisation of culture. The
insights would be strengthened by a more holistic study of different approaches
to operationalising culture.

# 6.2 Future research: culture and purpose

Throughout this discussion it has become clear that culture alone is only one ingredient of the recipe that leads to a (good) corporation of the future. Culture is closely connected to a number of areas, most closely to the question of the corporation's purpose, which is something that our interviewees highlighted (see also British Academy 2018). Some might want to argue that a specific culture can shape the way the corporation understands its role in society: that is, its social purpose. However, others emphasise that purpose is the foundation for culture. Salz, for example, emphasises that '[c]ulture ... gets determined by the way the group shares and acts upon its collective sense of purpose' (Salz 2013: 79). Furthermore, employee surveys demonstrate two crucial findings: first, that employees consider a workplace to be a 'good' one if they experience a sense of meaning or purpose in their work and, secondly, if the culture of the corporation is such that employees feel included and treated fairly. Whereas it might be easier to put these values into practice in environments of higher skilled work, it is particularly important to do so for lower skilled work environments as well. Research from the MGI shows that, even in environments which are traditionally considered to be particularly apt to award pay rises, the increase in remuneration becomes less important to employees vis-à-vis non-material values.

However, this kind of research shows only that there *is* some kind of influence between the purpose of a corporation, its culture (and employees' attitude to work). Based on this review, some immediate next steps to address in future research are thus:

- The direction and kind of influence between corporate culture and purpose: For instance, the exact relationship between corporate culture and purpose should be clarified. Is the achievement of an organisation's purpose dependent on its culture? What is the causal direction of fit between the two constructs?
- Possibility and potential success of a combined measurement framework: Insofar as both culture and purpose are inextricably linked, can they be measured in a combined framework? If so, how can they be measured in a combined framework?

- The role of leadership for the operationalisation of both corporate culture and influence: It seems clear that the leadership has *an* influence on corporate culture. However, the existing literature does not sufficiently discuss evidence of the importance of the leadership's focus on culture. Moreover, the discussion could profit significantly from an in-depth analysis of best-practice examples of how the leadership can implement culture change.
- The current research on both purpose and culture shows that the two are very closely linked. Hence, the field would benefit from a systematic study along the dimensions suggested here.

#### Acknowledgements

The authors would like to thank the participants of the British Academy's Roundtable discussions on the Future of the Corporation for their valuable input and challenges to an earlier draft of this article. Furthermore, the authors are particularly grateful to the speakers and discussants who participated at the *Future of the Corporation* Conference, held at the British Academy in May 2018. The contributions and discussions here, were of great help in further developing this paper.

Particular thanks are to be extended to the British Academy for giving us the opportunity to do this research and for the great collegial work throughout the research and writing process.

#### REFERENCES

Alvesson, M. (1993), Cultural Perspectives on Organizations (Cambridge, Cambridge University Press). Alvesson, M. (2002), Understanding Organizational Culture (London, SAGE).

https://doi.org/10.4135/9781446280072

Alvesson, M. & Berg, P. (1992), Corporate Culture and Organizational Symbolism (Berlin, Walter de Gruyter). https://doi.org/10.1515/9783110136074

Alvesson, M. & Sveningsson, S. (2008), Changing Organizational Culture: Cultural Change Work in *Progress* (New York, Routledge).

Arrow, K. J. (1974), The Limits of Organization (New York, W. W. Norton).

Ashkanasy, N. M., Wilderom, C. P. M., & Peterson, M. F. (eds) (2000), *Handbook of Organizational Culture and Climate* (Thousand Oaks, CA, SAGE).

Burnaz, S., Ataken, M. G. S., Topcu, Y. I. & Singhapakdi, A. (2010), 'An Exploratory Cross-cultural Analysis of Marketing Ethics: The Case of Turkish, Thai, and American Businesspeople', *Journal of Business Ethics*, 90: 371–82. https://doi.org/10.1007/s10551-010-0422-4

British Academy (2018), Roundtable Discussion on the Corporate Culture Workstream, British Academy, London.

Boyce, A., Nieminen, L. R. G., Gillespie, M. A., Ryan, A. M. & Denison, D. R. (2015), 'Which Comes First, Organizational Culture or Performance? A Longitudinal Study of Causal Priority with Automobile Dealerships', *Journal of Organizational Behavior*, 36: 339–59.

- https://doi.org/10.1002/job.1985
- Cameron, K. S., & Ettington, D. R. (1988), 'The Conceptual Foundations of Organizational Culture', in J. C. Smart (ed.) *Higher Education: Handbook of Theory and Research* Volume 4 (New York, Agathon), 356–96.
- Cameron, K.S. & Quinn, R. E. (1999), *Diagnosing and Changing Organizational Culture* (Reading, MA, Addison-Wesley).
- Chatman, J. A. & Jehn, K. A. (1994), 'Assessing the Relationship Between Different Levels of Leadership in Predicting Self- and Collective Efficacy: Evidence for Discontinuity', *Journal of Applied Psychology*, 87: 549–56.
- Cohen, P. N. (2013), 'The Persistence of Workplace Gender Segregation in the US', *Sociology Compass*, 7: 889–99. https://doi.org/10.1111/soc4.12083
- Coleman, J. (2013), 'Six Components of a Great Corporate Culture', *Havard Business Review*. https://hbr.org/2013/05/six-components-of-culture
- Crémer J. (1993), 'Corporate Culture and Shared Knowledge', *Industrial and Corporate Change*, 2: 351–86.
- Deal, T. E., & Kennedy, A. A. (1982), Corporate Cultures (Reading, MA, Addison-Wesley).
- Denison, D. R. (1990), Corporate Culture and Organizational Effectiveness (New York, Wiley).
- Denison, D. R. (1996), 'What is the Difference Between Organizational Culture and Organizational Climate? A Native's Point of View on a Decade of Paradigm Wars', *Academy of Management Review*, 21: 619–54. https://doi.org/10.5465/amr.1996.9702100310
- Denison. D. R. & Neale, W. S. (1996), Denison Organizational Culture Survey (Ann Arbor, MI, Aviat).
- Denison, D. R., Nieminen, L. & Kotrba, L. (2012), 'Diagnosing Organizational Cultures: A Conceptual and Empirical Review of Culture Effectiveness Surveys', *European Journal of Work and Organizational Psychology*, 23: 145–61. https://doi.org/10.1080/1359432X.2012.713173
- Desvaux, G., Devillard, S. & Sancier-Sultan, S. (2010), 'Women Matter, Gender diversity at the Top of Corporations: Making it Happen', *Women Matter 2010*, McKinsey & Company.
- Detert, J., Schroeder, R. & Mauriel, J. (2000), 'A Framework for Linking Culture and Improvement Initiatives in Organizations', *The Academy of Management Review*, 25: 850–63. https://doi.org/10.5465/amr.2000.3707740
- Devillard *et al.* (2013), 'Gender Diversity in Top Management: Moving Corporate Culture, Moving Boundaries', *Women Matter 2013*, McKinsey & Company.
- Douglas, P. C., Davidson, R. A. & Schwartz, B. N. (2001), 'The Effect of Organizational Culture and Ethical Orientation on Accountants' Ethical Judgments', *Journal of Business Ethics*, 34: 101–21. https://doi.org/10.1023/A:1012261900281
- Facebook (2018), 'Culture'. https://newsroom.fb.com/company-info/ [accessed: 8 May 2018].
- Feloni, R. (2016), 'The 5 Values Facebook Looks for in Every Employee', *Business Insider UK*. http://uk.businessinsider.com/facebook-core-values-2016-2?r=US&IR=T
- Fritzsche, D. J., & Becker, H. (1984), 'Linking Management Behavior to Ethical Philosophy', *Academy of Management Journal*, 27: 166–75.
- FSB (2017), 'Stocktake of Efforts to Strengthen Governance Frameworks to Mitigate Misconduct Risks', Financial Stability Board.

  http://www.fsb.org/2017/05/stocktake-of-efforts-to-strengthen-governance-frameworks-to-mitigate-misconduct-risks/
- Foust-Cummings, H. Prime, J., Salib, E. R. (2012), 'Calling All White Men: Can Training Help Create Inclusive Workplaces?', *Catalyst*. http://www.catalyst.org/knowledge/calling-all-white-men-can-training-help-create-inclusive-workplaces
- Goran, J, Srinivasan, R. & LaBerge, L. (2017), 'Culture for a Digital Age', *McKinsey Quarterly*, July. https://www.mckinsey.com/business-functions/digital-mckinsey/our-insights/culture-for-a-digital-age

- Gregory, B.T., Harris, S. G., Armenakis, A. A. & Shook, C. L. (2009), 'Organizational Culture and Effectiveness: A Study of Values, Attitudes, and Organizational Outcomes', *Journal of Business Research*, 62: 673–9. https://doi.org/10.1016/j.jbusres.2008.05.021
- Groysberg, B., Lee, J., Price, J. & Cheng, J. Y.-J. (2018), 'The Leaders Guide to Corporate Culture', *Harvard Business Review*. https://hbr.org/product/the-leaders-guide-to-corporate-culture/R1801B-PDF-ENG
- Halaj, Martin (2017), 'How to Measure the Safety Culture of Organizations', in *CBU International Conference Proceedings*. 5, 622–6. https://doi.org/10.12955/cbup.v5.996
- Hartnell, C.A., Ou, A. Y. & Kinicki A. (2011), 'Organizational Culture and Organizational Effectiveness: A Meta-analytic Investigation of the Competing Values Framework's Theoretical Suppositions', *Journal of Applied Psychology*, 96: 677–94. https://doi.org/10.1037/a0021987
- Hawes, C (2008), 'Representing Corporate Culture in China: Official, Academic and Corporate Perspectives', *The China Journal*, Number 59: 33–61. https://doi.org/10.1086/tcj.59.20066379
- Higgs, D. (2003), *Review of the Role and Effectiveness of Non-executive Directors*, Department of Trade and Industry, UK, http://www.ecgi.org/codes/documents/higgsreport.pdf [Last accessed: 05/05/2018].
- Huhtala, M., Feldt, T., Lamsa, A., Mauno, S. & Kinnunen, U. (2011), 'Does the Ethical Culture of Organisations Promote Managers' Occupational Well-being? Investigating Indirect Links via Ethical Strain', *Journal of Business Ethics*, 101: 231–47. https://doi.org/10.1007/s10551-010-0719-3
- Hunt, S. D., Wood, V. R. & Chonko, L. B. (1989), 'Corporate Ethical Values and Organizational Commitment in Marketing', *Journal of Marketing*, 53: 79–90. https://doi.org/10.1177/002224298905300309
- Ishak, W. (2017), 'Creating an innovation Culture', *McKinsey Quaterly*, September. https://www.mckinsey.com/business-functions/strategy-and-corporate-finance/our-insights/creating-an-innovation-culture
- Kaptein, M. (2008), 'Developing and Testing a Measure for The Ethical Culture of Organizations: The Corporate Ethical Virtues Model', *Journal of Organizational Behavior*, 29: 923–47. https://doi.org/10.1002/job.520
- Kaptein, M. (2011a), 'From Inaction to External Whistleblowing: The Influence of the Ethical Culture of Organizations on Employee Responses to Observed Wrongdoing', *Journal of Business Ethics*, 98: 513–30. https://doi.org/10.1007/s10551-010-0591-1
- Kaptein, M. (2011b), 'Understanding Unethical Behaviour by Unravelling Ethical Culture', *Human Relations*, 64: 843–69. https://doi.org/10.1177/0018726710390536
- Key, S. (1999), 'Organizational Ethical Culture: Real or Imagined?', *Journal of Business Ethics*, 20: 217–225. https://doi.org/10.1023/A:1006047421834
- Koh, H. C. & Boo, E. H. Y. (2001), 'The Link Between Organizational Ethics and Job Satisfaction: A Study of Managers in Singapore', *Journal of Business Ethics*, 29: 309–24. https://doi.org/10.1023/A:1010741519818
- Kreps, G. L. (1986), Organizational Communication: Theory and Practice (New York, Longman).
- Kotrba, L. M., Gillespie, M. A., Schmidt, A. M., Smerek, R. E., Ritchie, S. A. & Denison, D. R. (2012), 'Do Consistent Corporate Cultures Have Better Business Performance? Exploring the Interaction Effects', *Human Relations*, 65: 241–62. https://doi.org/10.1177/0018726711426352
- Lawrence, P. & Lorsch, J. (1967), 'Differentiation and Integration in Complex Organizations', *Administrative Science Quarterly*, 12: 1–30. https://doi.org/10.2307/2391211
- Lee, S. K. J. & Yu, K. (2004), 'Corporate Culture and Organizational Performance', *Journal of Managerial Psychology*, 19: 340–59. https://doi.org/10.1108/02683940410537927
- Lewin, K., Lippitt, R. & White, R. K. (1939), 'Patterns of Aggressive Behavior in Experimentally Created "Social Climates", *Journal of Social Psychology*, 10: 271–99. https://doi.org/10.1080/00224545.1939.9713366

- Lindell, M. & Brandt, C. (2000), 'Climate Quality and Climate Consensus as Mediators of the Relationship Between Organizational Antecedents and Outcomes', *Journal of Applied Psychology*, 85: 331–48. https://doi.org/10.1037/0021-9010.85.3.331
- Louis, M. R. (1980), 'Surprise and Sense Making: What Newcomers Experience in Entering Unfamiliar Organizational Settings', *Administrative Science Quarterly*, 25: 226–50. https://doi.org/10.2307/2392453
- Louis, M. R. (1985), 'An Investigator's Guide to Workplace Culture', in P. J. Frost, L. F. Moore, M. R. Louis, C. C. Lundberg & J. Martin *Organizational Culture* (Beverly Hills, CA, SAGE), 73–93.
- Lundberg, C. (1985), 'On the Feasibility of Cultural Intervention in Organizations', in P. J. Frost, L. F. Moore, M. R. Louis, C. C. Lundberg & J. Martin *Organizational Culture* (Beverly Hills, CA, SAGE), 169–85.
- Martin, J. (2002), *Organizational Culture: Mapping the Terrain* (Thousand Oaks, CA, SAGE). https://doi.org/10.4135/9781483328478
- Martin, J. & Siehl, C. (1983), 'Organizational Culture and Counterculture: An Uneasy Symbiosis', Organizational Dynamics, 12(2): 52–64. https://doi.org/10.1016/0090-2616(83)90033-5
- Mayer, D. (2014), 'A Review of the Literature on Ethical Climate and Culture', in B. Schneider & K. M. Barbera (eds) *The Oxford Handbook of Organizational Climate and Culture* (Oxford, Oxford University Press), 415–42.
- Meek, V. (1988), 'Organizational Culture: Origins and Weaknesses', *Organization Studies*, 9: 453–73. https://doi.org/10.1177/017084068800900401
- Michela, J. L. & Burke, W. W. (2000), 'Organizational Culture and Climate in Transformations for Quality and Innovation', in N. M. Ashkanasy, C. P. M. Wilderom & M. F. Peterson (eds) *Handbook of Organizational Culture and Climate* (Thousand Oaks, CA, SAGE), 225–44.
- Newcombe, T. (2013), 'Male Corporate Culture the Biggest Barrier to Women Reaching the Top', *HRmagazine.co.uk*.

  http://www.hrmagazine.co.uk/article-details/male-corporate-culture-the-biggest-barrier-to-women
  - reaching-the-top

    en S. A. (2018) 'Ulber is under a federal investigation for gender discrimination'. CNN Tech
- O' Brien, S. A. (2018) 'Uber is under a federal investigation for gender discrimination', *CNN Tech*. https://money.cnn.com/2018/07/16/technology/uber-gender-discrimination-investigation/index. html
- O'Reilly, C.A., Chatman, J. & Caldwell, D. F. (1991), 'People and Organizational Culture: A Profile Comparison Approach to Assessing Person–Organization Fit', *Academy of Management Journal*, 34: 487–516.
- Ostroff, C., Kinicki, A. J. & Muhammad, R. S. (2003), 'Organizational culture and climate', in N. W. Schmitt & S. Highhouse (eds) *Handbook of Psychology, Volume 12: Industrial and Organizational Psychology* 2nd edn (Hoboken, NJ, Wiley), 643–76. https://doi.org/10.1002/9781118133880.hop212024
- Ott, J. S. (1989), The Organizational Culture Perspective (Pacific Grove, CA, Brooks-Cole).
- Peters, T. J., & Waterman, R. H., Jr. (1982), In Search of Excellence (New York, Harper and Row).
- Peterson, D. (2002), 'Deviant Workplace Behavior and the Organization's Ethical Climate', *Journal of Business and Psychology*, 17(1): 47–61. https://doi.org/10.1023/A:1016296116093
- Pettigrew, A. M. (1979), 'On Studying Organizational Cultures', *Administrative Science Quarterly*, 24: 570–81. https://doi.org/10.2307/2392363
- Pettigrew, A. M. (1990), 'Organizational Climate and Culture: Two Constructs in Search of a Role', in B. Schneider (ed.) *Organizational Climate and Culture* (San Francisco, CA, Jossey-Bass), 413–34.
- Pierce, B. & Sweeney, B. (2010), 'The Relationship Between Demographic Variables and Ethical Decision Making of Trainee Accountants', *Internal Journal of Auditing*, 14: 79–99. https://doi.org/10.1111/j.1099-1123.2009.00404.x
- Prime, J. (2013) 'Creating an Inclusive Workplace—Starting with White Men', *Theguardian.co.uk*. https://www.theguardian.com/women-in-leadership/2013/apr/17/inclusive-workplace-starting-men

- Quinn, R. E. & Rohrbaugh, J. (1983), 'A Special Model of Effectiveness Criteria: Toward a Competing Values Approach to Organizational Analysis', *Management Science*, 29: 363–77. https://doi.org/10.1287/mnsc.29.3.363
- Rubin, R. S., Dierdorff, E. C. & Brown, M. E. (2010), 'Do Ethical Leaders Get Ahead? Exploring Ethical Leadership and Promotability', *Business Ethics Quarterly*, 20: 215–23. https://doi.org/10.5840/beq201020216
- Salz, A. (2013), Salz Review: An Independent Review of Barclay's Business Practices, Barclays plc.
- Savitz, A. (2013), The Triple Bottom Line: How Today's Best-run Companies Are Achieving Economic, Social and Environmental Success—and How You Can Too (San Francisco, CA, Jossey-Bass).
- Schaubroeck, J. M., Hannah, S. T., Avolio, B. J., Kozlowski, S. W., Lord, R. G., Treviño, L. K., Dimotakis, N. & Peng, A C. (2012), 'Embedding Ethical Leadership Within and Across Organizational Levels', *Academy of Management Journal*, 55: 1053–78. https://doi.org/10.5465/amj.2011.0064
- Schein, E. H. (1985), Organizational Culture and Leadership (San Francisco, CA, Jossey-Bass).
- Schein, E. H. (1991), 'What is Culture?', in P. J. Frost, L. F. Moore, M. R. Louis, C. C. Lundberg & J. Martin (eds) *Reframing Organizational Culture* (Newbury Park, CA, SAGE), 243–53.
- Schein, E. H. (1992), Organizational Culture and Leadership, 2nd edn (San Francisco, CA, Jossey Bass.)
- Schein, E. H. (2010), Organizational Culture and Leadership 4th edn (San Francisco, CA, Jossey-Bass).
- Schein, E. H., Schein, P. (2016), Organizational Culture and Leadership 5th edn (New York, Wiley).
- Schlicht, E. (2004), 'Social Evolution, Corporate Culture, and Exploitation', *Journal of Institutional and Theoretical Economics (JITE) | Zeitschrift Für Die Gesamte Staatswissenschaft*, 160(2): 232–42. https://doi.org/10.1628/0932456041438777
- Schneider, B., & Reichers, A. E. (1983), 'On the Etiology of Climates', *Personnel Psychology*, 36: 19–39. https://doi.org/10.1111/j.1744-6570.1983.tb00500.x
- Schneider, B., White, S. S. & Paul, M. C. (1998), 'Linking Service Climate and Customer Perceptions of Service Quality: Test of a Causal Model', *Journal of Applied Psychology*, 83: 150–63. https://doi.org/10.1037/0021-9010.83.2.150
- Schneider, B., Erhart, M. G. & Macey, W. A. (2011), 'Perspectives on Organizational Climate and Culture', in S. Zedeck (ed.) *APA Handbook of Industrial and Organizational Psychology: Volume 1. Building and Developing the Organization* (Washington, DC, American Psychological Association), 373–414. https://doi.org/10.1037/12169-012
- Schwartz, H. & Davies, S. (1981), 'Matching Corporate Culture and Business Strategy', *Organizational Dynamics*, 10(1): 30–8. https://doi.org/10.1016/0090-2616(81)90010-3
- Shih, C. & Chen, C. (2006), 'The Effect of Organizational Ethical Culture on Marketing Managers' Role Stress and Ethical Behavioral Intentions', *The Journal of American Academy of Business*, 8: 89–95.
- Solomon, R. C. (1992), Ethics and Excellence (New York, Oxford University Press).
- Solomon, R. C. (1999), A Better Way to Think About Business: How Personal Integrity Leads to Corporate Success (New York, Oxford University Press.
- Solomon, R. C. (2000), 'Business with Virtue: Maybe Next Year', *Business Ethics Quarterly*, 10: 319–31. https://doi.org/10.2307/3857718
- Solomon, R. C. (2004), 'Aristotle, Ethics and Business Organizations', *Organization Studies*, 25: 1021–43. https://doi.org/10.1177/0170840604042409
- Sweeney, B., Arnold, D. & Pierce, B. (2010), 'The Impact of Perceived Ethical Culture of the Firm and Demographic Variables on Auditors' Ethical Evaluation and Intention to Act Decisions', *Journal of Business Ethics*, 93: 531–51. https://doi.org/10.1007/s10551-009-0237-3
- Thompson, J. D. (1967), Organization in Action (New York, McGraw-Hill).
- Tichy, N. M. (1982), 'Managing Change Strategically: The Technical, Political, and Cultural Keys', Organizational Dynamics, 11(2): 59–80. https://doi.org/10.1016/0090-2616(82)90005-5

- Treviño, L. K., & Butterfield, K. D. & McCabe, D. L. (1998), 'The Ethical Context in Organizations: Influences on Employee Attitudes and Behaviours', *Business Ethics Quarterly*, 8: 447–76. https://doi.org/10.1007/s10551-009-0237-3
- Treviño, L. K., Weaver, G. R., & Reynolds, S. J. (2006), 'Behavioural Ethics in Organizations: A Review', Journal of Management, 32: 951–90 https://doi.org/10.1177/0149206306294258
- Trice, H. M. & Beyer, J. M. (1993), *The Cultures of Work Organizations* (Englewood Cliffs, NJ: Prentice-Hall).
- Tsai, M. & Shih, C. (2005), 'The Influences of Organizational and Personal Ethics on Role Conflict Among Marketing Managers: An Empirical Investigation', *International Journal of Management*, 22: 54–61.
- Turnstall, W. B. (1983), 'Cultural Transition at AT&T', Sloan Management Review, 25(1): 1-12.
- Verbeke, W., Volgering, M. & Hessels, M. (1998), 'Exploring the Conceptual Expansion Within the Field of Organizational Behaviour: Organizational Climate and Organizational Culture', *Journal of Management Studies*, 35(3): 303–29. https://doi.org/10.1111/1467-6486.00095
- Victor, B. & Cullen, J. B. (1987), 'A Theory and Measure of Ethical Climate in Organizations', *Research in Corporate Social Performance and Policy*, 9: 51–71.
- Victor, B. & Cullen, J. B. (1988), 'The Organizational Bases of Ethical Work Climates', *Administrative Science Quarterly*, 33: 101–25. https://doi.org/10.2307/2392857
- Webb, W. N. (2012), 'Ethical Culture and the Value-based Approach to Integrity Management: A Case Study of the Department of Correctional Services', *Public Administration and Development*, 32: 86–108. https://doi.org/10.1002/pad.1602
- Williams, B. (1985), Ethics and the limits of philosophy. Cambridge, MA: Harvard University Press.
- Williamson, I. O., Burnett, M. F. & Bartol, K. M. (2009) 'The Interactive Effect of Collectivism and Organizational Rewards on Affective Organizational Commitment', *Cross Cultural Management:*An International Journal, 16(1): 28–43. https://doi.org/10.1108/13527600910930022
- Wimbush, J., Shepard, J. M. & Markham, S. E. (1997), 'An Empirical Examination of the Multi-dimensionality of Ethical Climate in Organizations,' *Journal of Business Ethics*, 16(1): 67–77. https://doi.org/10.1023/A:1017962229330
- Wittayapoom, K. & Limsuwan, S. (2012), 'How Does Internal Control Effectiveness Create Reliability of Financial Reporting? An Empirical Research of the Thai Listed Firms', *Journal of International Business and Economics*, 12: 80–96.
- Zhang, J., Chiu, R. & Wei, L. (2009a), 'Decision-making Process of Internal Whistleblowing Behavior in China: Empirical Evidence and Implications', *Journal of Business Ethics*, 88: 5–41. https://doi.org/10.1007/s10551-008-9831-z
- Zhang, J., Chiu, R. & Wei, L. (2009b), 'On Whistleblowing Judgment and Intention: The Roles of Positive Mood and Organizational Ethical Culture', *Journal of Managerial Psychology*, 24: 627–49. https://doi.org/10.1108/02683940910989020
- Zohar, D. & Hofmann, D. A. (2012), 'Organizational Culture and Climate', in S. W. . Kozlowski (ed.) *The Oxford Handbook of Industrial and Organizational Psychology* (Oxford, Oxford University Press), 643–66. https://doi.org/10.1093/oxfordhb/9780199928309.013.0020

#### *Note on the authors:*

Nien-hê Hsieh is an Associate Professor of Business Administration in the General Management Unit of Harvard Business School. His research concerns ethical issues in business and the responsibilities of global business leaders. nine-he@principia-advisory.com

Benjamin Lange is a D.Phil. Philosophy candidate at St. Anne's College at the University of Oxford. His research focuses on issues in ethics and political philosophy. benjamin.lange@philosophy.ox.ac.uk

David Rodin is Part Time Professor at the European University Institute in Florence and CEO of Principia Advisory, a consultancy specialising in organisational ethics and culture change.

david@principia-advisory.com

Mira L. A. Wolf-Bauwens is a Visiting Scholar at Columbia University in the City of New York, as part of her PhD in Political Philosophy at the University of Zurich. Her research centres on democratic and recognition theory, with a particular focus on the relationship between citizens and institutions. mira.wolf-bauwens@uzh.ch

To cite the article: Nien-hê Hsieh, Benjamin Lange, David Rodin and Mira L.A. Wolf-Bauwens (2018), 'Getting clear on corporate culture: conceptualisation, measurement and operationalisation', *Journal of the British Academy*, 6(s1): 155–184. DOI https://doi.org/10.5871/jba/006s1.155

This article is licensed under a Creative Commons Attribution-NonCommercial-NoDerivs 4.0 Unported License.

Journal of the British Academy (ISSN 2052–7217) is published by The British Academy—the national academy for the humanities and social sciences. 10–11 Carlton House Terrace, London, SW1Y 5AH www.britac.ac.uk